Massachusetts Institute of Technology Handout 20 6.042J/18.062J: Mathematics for Computer Science October 6, 2000 Professors Nancy Lynch and Srinivas Devadas

Mini-Quiz 4

1. Write your name:

2. Describe an algorithm that locates the first occurrence of the largest element in a finite list of integers, where the integers are not necessarily distinct.

Solution:

procedure first largest(a_1, . . . , a_n : integers) max := a_1; location := 1; for i := 2 to n

if max < a_i then max := a_i; location := i; return location;